Author's response to reviews

Title: Retreatment of hepatitis C patients with pegylated interferon combined with ribavirin in non-responders to interferon plus ribavirin. Is it different in real life?

Authors:

Fernando L Gonçales Jr (flgj@uol.com.br)
Camila A Moma (cami_atm@yahoo.com.br)
Aline G Vigani (aline.vigani@sigmanet.com.br)
Eduardo SL Gonçales (dugoncales@hotmail.com)
Adriana FF Angerami (afcfeltrin@yahoo.com.br)
Raquel Tozo (tumoca@uol.com.br)
Maria Helena P Pavan (mariahpavan@uol.com.br)
Neiva SL Gonçales (neiva@unicamp.br)

Version: 4 Date: 5 July 2010

Author's response to reviews:

Dear editor

I am sending a new version of the manuscript “Retreatment of hepatitis C patients with pegylated interferon combined with ribavirin in non-responders to interferon plus ribavirin. Is it different in real life?”

I am also sending the response to the reviewer.

Thank you very much for your attention

Fernando L Gonçales Jr

Date: 04 July 2010

Dear referee,

In relation to the English, the text was revised by a native speaker resident in Brazil.

In the last submission we sent the manuscript with the abstract. We don’t know what happened.

As you suggested we reported the SVR rates as per-protocol analysis as well as intention-to-treat-analysis in the results, abstract and discussion.

We appreciate the attention and cooperation given to our work and all suggestions.

Best Regards,

Fernando L Gonçales Jr
4 July 2010